Author’s response to reviews

Title: Prevalence and evolution of intimate partner violence before and during pregnancy: a cross-sectional study

Authors:

An-Sofie Van Parys (ansofie.vanparys@ugent.be)
Ellen Deschepper (ellen.deschepper@UGent.be)
Kristien Michielsen (kristien.michielsen@ugent.be)
Marleen Temmerman (marleen.temmerman@ugent.be)
Hans Verstraelen (hans.verstraelen@ugent.be)

Version: 2
Date: 28 March 2014

Author’s response to reviews:

in reply to your email dd March 18th, I made the requested adaptations:
- added a competing interest and authors contributions section
- placed the tables after the reference section

Hopefully this will be sufficient to start the peer review process.

Kind regards